Spatial Heterogeneity and Population Mobility in India

Jajati Keshari Parida S Madheswaran ISBN 81-7791-190-2

© 2010, Copyright Reserved The Institute for Social and Economic Change, Bangalore

Institute for Social and Economic Change (ISEC) is engaged in interdisciplinary research in analytical and applied areas of the social sciences, encompassing diverse aspects of development. ISEC works with central, state and local governments as well as international agencies by undertaking systematic studies of resource potential, identifying factors influencing growth and examining measures for reducing poverty. The thrust areas of research include state and local economic policies, issues relating to sociological and demographic transition, environmental issues and fiscal, administrative and political decentralization and governance. It pursues fruitful contacts with other institutions and scholars devoted to social science research through collaborative research programmes, seminars, etc.

The Working Paper Series provides an opportunity for ISEC faculty, visiting fellows and PhD scholars to discuss their ideas and research work before publication and to get feedback from their peer group. Papers selected for publication in the series present empirical analyses and generally deal with wider issues of public policy at a sectoral, regional or national level. These working papers undergo review but typically do not present final research results, and constitute works in progress.

# SPATIAL HETEROGENEITY AND POPULATION MOBILITY IN INDIA

# Jajati Keshari Parida<sup>\*</sup> and S Madheswaran<sup>\*\*</sup>

### Abstract

Mobility is one of the important aspects of human nature, which is often guided by socioeconomic, political as well as environmental factors. The nature, pattern and direction of population mobility may vary across the space. The dynamics of internal migration in India plays an important role in the process of economic development and social transformation and shows an increasing trend of rural to urban flow over the years. At the same time, it shows falling trends in all other streams of migration and are registering negative growth rates as well. The determinants of rural to urban migration include a set of socio-economic, demographic, geographical and environmental variables. The empirical results establish the "Gravity Model" of migration in India; where as the "Harris-Todaro Model" of rural urban migration has limited applicability in both inter-state and intra-state migration in India.

#### 1. Introduction

The concept of internal migration now connotes much wider dimension that varies from daily commuting to nearby places on one hand, to permanent shift of residence to distant places on the other. The nature and pattern of migration varies from one social group of migrants to another because of the fact that the rural migrants are not a homogenous group (Mitra and Murayama, 2008). Rural to urban migration has attracted the attention of academicians as well as the policymakers throughout the world in recent years, because of its wide range of socio-economic, political, demographic, ecological and environmental implications. It plays an important role in the process of economic development and social transformation because the shifting of the work force from primary to secondary and tertiary sectors that result in structural change. The improvement in infrastructure and communication accompanying economic development could result in increased population mobility. But from the earlier studies (Kundu and Gupta, 1996; Singh, 1998; and Srivastava, 1998), it is evident that migration as a percentage of total population has been declining up to 1991 census, where as the rural to urban migration is showing an increasing trend over the period 1971 to 2001(Lusome & Bhagat, 2006) The recent report of United Nations Development Programme (UNDP, 2009) entitled 'Mobility and Human Development' stated that population shift from villages to cities is natural and should be encouraged and that it results in concentration of production, mobility of people and economic integration to lift rural people out of poverty in India. Hence, any long-term plan aiming at an increase in labour productivity and rural employment opportunities should consider the possibility of migration of a relatively large proportion of population away from areas with relatively low agronomic potential (Dhar, 1980; and Kundu & Sarangi, 2007). The availability of good agricultural land continues to be one of the most powerful economic factors determining the magnitude and direction of population migration. The

<sup>\*</sup> Research Scholar, Centre for Economic Studies and Policy, ISEC, Bangalore.

<sup>\*\*</sup> Professor, Centre for Economic Studies and Policy, ISEC, Bangalore.

We are thankful to Prof Abdul Ajiz, Prof KS James, Prof R S Deshpande, Prof M R Narayana, Smt B P Vani, Prof Meenakshi Rajeev and Dr K Gayithri for their valuable suggestions. Thanks are also due to the unknown referees for their thought provoking comments and suggestions.

agricultural unemployment and underemployment, lower agricultural wage and poverty are the major factors pushing labourers towards job opportunities in urban areas, while the availability of employment opportunities, relatively higher wages, better educational opportunities, better availability of transportation and infrastructure etc. are the major factors in the rural-urban migration (Harris and Todaro, 1970; Barro & Sala-i-Martin, 1991; Gupta, 1993; Bhattacharya, 1998; Pekkala and Tervo, 2002; Bhattacharya, 2002; and Joshi & Lobo, 2003; Andrienko and Sergei, 2003). Therefore, in this context, an attempt has been made to examine the broad trends and patterns of internal migration and to study the determinants of rural to urban migration in India.

The paper is organised as follows. Section 2 provides a brief review of literature. Section 3 gives the data sources used in the present study both for the estimation of trends and patterns, and determinants of migration. Section 4 analyses the broad trends and patterns of internal migration and the intercensal growth of migrants for three intercensal periods as well as provides the reasons for migration. The empirical model for the estimation of migration function is given in section 5. Section 6 provides the empirical results and section 7 presents the conclusion.

### 2. A Brief Review of Literature

Internal migration is defined as 'population movement across a political or administrative boundary, which entails a change of usual residence' (Bilsborrow, 1998). The migration literature provides a diverse range of related definitions based on three dimensions, which are explained by Kothari (2002). The first dimension entails the motives for migrations. The reasons to migrate are in many cases not just economic, social, political and cultural aspects also play its part. A migrant from an impoverished backward region may be attracted to more prosperous regions because of better public services, higher wages, more business opportunities, more employment opportunities etc. The second dimension is the geographical aspect of population movement. Migration flows form a certain spatial pattern, which depends on the distance the migrant's travel and the direction of their movements. The last dimension is time, which is for how long does the migrant stay in the destination area and how often does he or she travel between different destinations. The Indian Population Census classifies a person as a migrant if either (a) 'his/her place of birth was different from the place of enumeration' or (b) 'his/her place of last continuous residence was different from the place of enumeration'. Hence, the census provided data on both lifetime (birthplace) migration and last-move migration. The former definition is less practical because of the fact that it defines a person as a migrant even if he or she stays in the new destination area permanently. The Population Census of India collected information regarding internal migration flows at three levels of spatial aggregation, i.e., migration flows between Indian states (interstate migration), migration flows between districts (inter-district migration) and migration flows within a district (Intra-district).

The concept of internal migration (from rural to urban) attracted the attention of both academicians and policymakers only after the seminal work of Lewis (1954). The inter-sectoral allocation of labour is the centerpiece in the dual economy analysis of Lewis (1954) and subsequent works by Ranis and Fei (1961). The main message of these studies is that in the process of development, labour moves to the modern sector which facilitates development. However, in developing

this idea, it is assumed that the modern sector faces perfectly elastic labour supply originating in the traditional or rural sector (i.e., there is unlimited supply of labour from rural to urban sectors). This view is inconsistent with the idea that migration is determined by varying income differentials and that labour is productive in all sectors of the economy. But labour migration from rural to urban sector increases labour productivity and hence it is always desirable and should be encouraged. Ranis and Fei (1961) are of the opinion that technological progress in agriculture sector will make migration process slow by increasing agricultural labour productivity. The Todaro (1969) and Harris and Todaro (1970) models explain the migration flows in the presence of rising unemployment rates in the destination (urban) region which in turn results in the development of informal sector. The internal migration is driven by the higher expected wage in urban sector. The expected wage is nothing but the product of higher urban (institutionally fixed) wage and probability of finding a job (calculated by dividing total availability of urban jobs and total urban labour force) in the urban sector. Further developments on micro modeling approach are 'the new economics of migration' which focus on family as the agent that maximize the family utility function by minimising the risks associated with the agricultural sector (Stark & Bloom 1985; Stark & Katz 1986; Banerjee, 1998; Taylor & Martin 2001; Stark 1991; Mincer, 1978). Similarly, the networks created by migrants in the destination region reduce the migration costs and hence encourages new migration flows to the destination places (Bhattacharya, 1998; Mitra, 2003).

Internal migration at macro level studies the relationship between aggregate migration flows and the presence of spatial heterogeneity among the different locations. The geographical differentials in terms of economic growth and development, is the main determinant of internal migration. The Gravity Model is the most common theoretical framework used in empirical analysis to study the spatial determinants of migrations. It argues that migration is directly correlated with population size and inversely correlated with the distance between the origin and the destination regions. Distance is a key variable, the proxy for all the migration costs, both psychological and monetary, that is spatially related to the sending and destination region. The population size is the sign of the rate of urbanisation or the growth of urban sector (Greenwood, 1985; Greenwood and Hunt, 2003; Larson and Mundlack, 1995). However, not all people react in the same way to differences between places. The selectivity influence of migration characterises the different propensity to migrate for different categories of people. Young people in the working age have a higher propensity to migrate than people in other age groups. Moreover, education is also an important selective factor. High skilled people tend to migrate more than low skilled people (Greenwood, 1997).

The majority of empirical works tries to investigate the impact of some economic variables on internal migration. A high economic prosperity also means more activities, services and opportunities for people living in that area. Moreover, dynamic centres attract mostly young people, who are widely recognised to be highly mobile. The most representative economic variable is the per capita gross domestic product (GDP) as explained by Greenwood (1997). Internal migration has greater potential for poverty reduction, meeting the Millennium Development Goals (MDGs) and contributing to economic growth in developing countries than international migration (Deshingkar, 2004, 2006; Deshingkar and Start, 2003; Srivastava and Sasikumar, 2004; Deshingkar et al, 2006). The relatively poor and backward states show large population mobility, which is primarily in search of livelihood. The mobility of male

population is also found to be prominent in the relatively advanced states like Maharashtra and Gujarat as explained by Mitra and Murayama (2008). The labour market variables, which are often included as explanatory variables in migration analysis, are the unemployment rate and wage rate. The increasing level of education for rural residents unaccompanied by increased opportunities for higher earnings implies an increase in the gap between rural earnings and hence a cause of migration (Diehl, 1966; Barkley, 1990; Perloff et al, 1998 and Agesa, 2001; Waddington and Sabates-Wheeler, 2004). The flow of migration to the major cities is the result of rural - urban dichotomies in income, employment opportunity and absorptive capacity as stated by Ullah (2004). Hossain (2001) found that while poverty, job searching and family influence were the main push factors for out-migration, better opportunity, prior migrants and availability of jobs were the main factors behind migration. The reason why people decide to move from one region to another is not just related to economic factors. A host of other factors play a role as well. The group of variables that can affect internal migration flows is guite broad and is related with the quality of life. All these factors concern public safety, social services, environmental quality, as well as political factors (Adrienko and Guriev, 2003). According to Adrienko and Guriev (2003), people move from poorer and job scarce regions with worse public good provision to areas that are richer and more prospering both in terms of employment prospects and public goods.

In this context, an attempt has been made to study (at the macro level) the trends, patterns and determinants of internal migration in India using the grouped data from Indian Census. To study the determinants of internal migration, a "Gravity Model" is used as the theoretical background for the empirical estimation. In the "Gravity Model", a set of other explanatory variables (besides population size and distance) are included to test the applicability of "Harris-Todaro Model" in both inter-state and intra-state migration in India.

### 3. Data Sources

The paper seeks to address two main questions, the first being the current patterns and trends of internal (both inter-state and intra-state) migration in India and the second one: 'What are the determinants of rural to urban migration in India?' The first question is analysed using the Census data. The migration tables (Census of India) are used to analyse the trends and broad patterns of internal migration at inter-state, inter-district and intra-district levels and for different sex categories. The intercensal growth of migrants is also estimated using the same migration tables. The data analysis for the study of determinants of migration is limited to males (Census of India, migration tables), since a large proportion of female migration in India is for non-economic reasons such as marriage. The data for the explanatory variables are taken from other sources like Ministry of Agriculture, Government of India (GoI); Lok Sabha Unstarred Question No. 193, 15<sup>th</sup> July, 2002; Department of Secondary and Higher Education, Ministry of Human Resource Development, GOI; Wage Rate in Rural India, Labour Bureau and Annual Survey of Industries, 2000-2001, Statistics on Employment and Labour, Ministry of labour, GOI; Central Statistical Organisation (CSO). Most of the data are collected from the website (http://www.mapsofindia.com).

# 4. Trends, Patterns and Intercensal Growth of Migration in India

#### 4.1. Trends of Internal Migration:

The total (in millions) and percentage of lifetime migrants based on the place of last residence criterion, by sex and type of residence in India from 1971 to 2001 is given in Table 1 and 2, respectively. In 1971 census, 159.6 million people, comprising of 49.6 million males and 110 million females, were termed as migrants on the basis of place of last residence criterion. This constitutes 30.6 per cent of the total population of the country. In terms of total volume of migration, the figure increased to 201.6 million in 1981, 225.9 million in 1991 and 309.4 million in 2001.

Year	L	ifetime Migrants (in millio	n)
i cai	Total	Males	Females
1971	159.6	49.6	110.0
1981	201.6	59.2	142.4
1991	225.9	61.1	164.8
2001	309.4	90.7	218.7

Table 1: Internal Migrants by Sex, India (1971-2001)

Source: Calculated from Census of India, Migration Tables.

The percentages of migrants to total population however declined to 30.3 in 1981 and further to 27.4 in 1991. It could be suspected that Census of India underestimates the total number of migrants. It is partially recurring seasonal migration and commuting (in the less than one year duration of last residence), which is the most significant emerging pattern of internal migration in India. The percentage of migrants is however increased to 30.07 per cent in 2001 and this increase may be due to the rapid expansion of the urban areas in India. The percentages of male migrants declined from 18.9 in 1971 to 17.8 in 1981 and further to 14.6 in 1991. It has however increased to 17.03 per cent in 2001. The phenomenal increase in the urban informal sector, particularly after the economic reforms (1991), could be responsible for this. The percentages of female migration increased to 48.3 in 1981 as compared to 42.3 in 1971 and in 1991, it was 41.2 per cent which again increased to 44.05 per cent in 2001. It is obviously due to the social-cultural setup in India where females normally migrate after marriage. In comparison to the rural and urban categories, the percentage of migrants in rural areas increased from 28.2 per cent in 1971 to 17.3 per cent in 2001. In both rural and urban areas, the share of female migrants is again found to be dominant.

Type of Residence	Year	Total	Male	Female
	1971	30.6	18.9	42.8
Total	1981	30.3	17.8	44.3
TOLAI	1991	27.4	14.6	41.2
	2001	30.07	17.03	44.05
	1971	28.2	14.1	43.1
Dural	1981	28.2	16.1	46.5
Rural	1991	26.8	13.4	41.8
	2001	30.09	13.2	47.9
	1971	28.2	14.1	43.1
L luis e a	1981	28.2	16.1	46.5
Urban	1991	26.8	13.4	41.8
	2001	17.3	13.9	21.03

Table 2: Percentage of Lifetime Migrants in the Total Population by Sex and Type of Residence in India, (1971-2001).

Source: Calculated from Census of India, Migration Tables.

# 4.2. Reasons for Migration:

Table 3: Reasons for Internal Migration according to Sex and different Streams of Migration in 2001 Census (figures in %)

	Male				Female			
Reasons for Migration	Rura⊦ Rural	Rura⊦ Urban	Urban- Rural	Urban- Urban	Rural- Rural	Rural- Urban	Urban- Rural	Urban- Urban
Work or Employment	28.87	26.51	50.64	38.53	2.17	3.5	5.0	4.1
Business	2.15	2.44	3.63	3.55	0.2	0.4	0.35	0.45
Education	6.27	5.16	7.08	6.16	0.71	1.56	2.85	2.61
Marriage	4.26	1.58	0.83	0.84	78.17	47.58	43.03	39.87
Moved after birth	14.46	24.19	5.11	9.01	3.66	14.92	4.66	6.92
Moved with Household	25.17	24.93	22.68	30.12	9.71	24.45	36.62	38.66
Other reasons	18.82	15.19	10.03	11.79	5.38	7.59	7.49	7.39
Total	100	100	100	100	100	100	100	100

Source: Calculated from Census of India, Migration Tables.

The reasons for migration have been classified into seven broad groups (viz., work/employment, business, education, marriage, moved at birth, moved with family and others) in 2001 Census. The data on reasons for migration are useful to understand the motivational factors behind the movement of people. It is observed from Table 3 that employment among males and marriage among females are the main reasons for migration. About 29 per cent of the male migrants moved form rural to rural areas due to work or employment reasons. Similarly, employment or work caused around 26.5 per cent from rural to urban, 50.64 per cent from urban to rural and 38.53 per cent from urban to urban migration in India. While in the case of females, 78.17 per cent migrated from

rural to rural areas due to marriage. Again 47.58 per cent from rural to urban, 43.03 per cent from urban to rural and 39.87 per cent from urban to urban females migrated due to marriage. Associational reasons like movement on account of accompanying parents or any other members of the family is regarded as the second most important reason among both male and female intercensal migrants.

# 4.3. Patterns of Internal Migration:

The total migration flow can broadly be divided into four different streams viz., rural to rural, rural to urban, urban to urban and urban to rural. Similarly from the point of view of distance, it can be classified as inter-state and intra-state (inter-district & intra-district) migration. Table 4 depicts the percentage distribution of lifetime migrants in different streams for males and females from 1971 to 2001 at intra-district, inter-district, inter-state distance categories, respectively. The results show a considerable decline in the proportion of intra-district migrants for both the sex categories, where as both inter-district and inter-state migrations are increasing over the same period of time. As explained by Bhagat (2005), droughts and floods which occurs in some parts of the country every year renders many people homeless. This has been recognised as one of the reasons for migration in the censuses of 1981 and 1991." And at the same time, the phenomenal expansion of urban informal sector during the 1990s could be the reason for migration in 2001 Census. At the national level, rural to rural migration constitutes the most dominant category. But its share has been declining over time for both males and females in all the three streams. The decline was greater in case of males as compared to females. There is a large share of female intra-district rural-to-rural migration, which is normally explained in terms of marriage migration (Census of India). Rural-to-rural intra-district migration of males seems to be due to their migration from areas of low agricultural productivity to sparsely populated areas with new developmental activities. In addition, Lusome & Bhagat, (2006) stated that apart from employment among males and marriage among females, moved with household emerged as another important factor for migration among males as well as females. The rural to urban streams show increased percentage share of migrants over the period, 1971 to 2001, which is mainly on account of employment or work.

Combining all the above three distance categories of migration (i.e., intra-district, inter-district and inter-state migration), it is suggested that rural-to-urban male migration has increased substantially over the period, 1971 to 2001. The creation of modern sector in major metropolises and big cities and the subsequent development of the urban informal sector as is the major reason for rural to urban migration. A similar trend is found for urban-to-urban migration. However, the proportion of rural-tourban lifetime migration of females has decreased with increasing distance between origin and destination places. The share of urban-to-urban migration of both males and females was comparatively low in the intra-district stream, but it increased substantially in the inter-district and inter-state streams. The reason could be that the institutions of higher learning, particularly professional and technical institutions, are not available in all districts and the motives for higher education drives both the urban dwellers as well as the rural folks to migrate over long distances.

Minnetien Oeteensiee		Males				Females			
Migration Categories	1971	1981	1991	2001	1971	1981	1991	2001	
	Intra-district Level								
Rural to Rural	36.8	32.3	30.9	29.9	61.3	56.6	54.7	54.6	
Rural to Urban	9.8	10.7	11.5	10.3	5.3	6.0	6.3	5.9	
Urban to Urban	3.6	3.6	3.6	4.6	3.0	3.0	2.9	2.7	
Urban to Rural	3.8	4.6	4.3	5.8	1.8	2.5	2.3	3.7	
Sub Total	54	51.2	50.3	50.6	71.4	68.1	66.2	66.9	
		Int	er-distric	t Level					
Rural to Rural	9.5	9.3	8.9	8.2	12.2	13.4	14.0	12.8	
Rural to Urban	8.0	9.6	10.5	8.9	3.2	3.9	4.5	4.4	
Urban to Urban	2.4	2.6	2.5	3.5	1.5	1.8	1.9	1.9	
Urban to Rural	7.2	8.4	8.0	7.2	3.4	4.2	4.2	3.9	
Sub Total	27.1	29.9	29.9	27.8	20.3	23.3	24.6	23.0	
		In	ter-state	Level					
Rural to Rural	4.5	3.8	3.5	3.4	3.6	3.3	3.3	3.8	
Rural to Urban	6.7	7.5	8.2	9.8	1.7	2.2	2.5	2.9	
Urban to Urban	1.5	1.4	1.8	2.7	0.7	0.6	0.7	0.4	
Urban to Rural	6.2	6.2	6.5	5.7	2.3	2.5	2.7	2.5	
Sub Total	18.9	18.9	19.8	21.6	8.3	8.6	9.2	10.1	
Grand Total	100	100	100	100	100	100	100	100	
		All di	stance ca	tegories					
Rural to Rural	50.8	45.2	43.3	28.8	76.8	73.3	72.0	66.6	
Rural to Urban	24.0	27.7	29.8	27.0	10.2	12.0	13.2	12.4	
Urban to Urban	7.5	7.6	7.4	5.0	5.1	5.5	5.4	3.9	
Urban to Rural	16.6	19.2	18.9	18.2	7.5	9.1	9.0	9.2	

Table 4: Internal Migration flows by Sex and different distance categories in India, 1971-2001 (figures in %).

Source: Calculated from Census of India, Migration Tables.

# 4.4. Intercensal Growth of Migrants in India:

The percentage growth of intercensal (1971-81, 1981-91 and 1991-2001 periods) migrants for all categories are presented in intra-district, inter-district, inter-state streams and the combination of all streams in Table 5. During 1971-81, there was a positive growth of total and female migrants for all migration categories while a negative growth of male migration from rural-rural areas in both intra-district and inter-state cases. Similarly, the growth rate of urban-urban male migration in inter-state cases is negative in the same period. Again during the decade 1981-91, both male and female migration decreased as registered by the negative growth rates. The growth rate of male migrants in all streams is negative except rural-urban inter-district and inter-state cases, while the growth rate of female migration from urban to rural areas is negative except in inter-state case. The resulting growth of total migration is positive in rural-urban areas and is negative in all other cases. The growth rate of intercensal migrants during 1991-01, for both male and female migrants is positive except urban-urban category in inter-district stream and rural-rural and rural to urban categories in intra-district male cases.

Male inter-state rural to urban migration stream shows an increase of nearly 90 per cent while females increased by about 61 per cent. On the other hand, inter-state urban to urban migration stream for males shows an increase of 28 per cent while females have increased by 22 per cent. These increases in rural - urban and urban - urban cases are greater than the increases in other streams of migration for both male and female categories. The main reasons for the above migration streams could be explained in terms of employment and marriage. Associational reasons including movement on account of accompanying parents or any other members of the family and for learning and higher education are considered t to be the second most important reason for both male and female migration. The reasons for the negative growth rates are mostly because of the inability of Indian Census to collect information regarding both seasonal migrants and commuters, which are the two emerging migration patterns in India during the last two decades. The Census of India collects the information at a particular point of time and there are no such questions in the Census questionnaire which can collect information regarding the seasonal migration and commuting. The reasons for the positive growth rate could be due to the creation of new districts or states during the intercensal period. The number of internal migrants could be either overstated or under estimated depending upon the creation of new districts or states during the intercensal period. Recently, three new states viz., Uttaranchal, Jharkhand and Chhattisgarh were created in 2000- a few months before the latest census held in February 2001.

Intercensal Growth Rates								
	1971-8	1		1981-9	91		1991-0	)1
Total	Males	Females	Total	Males	Females	Total	Males	Females
1		Intr	a-distric	t Level				
6.2	-1.5	9.4	-0.73	-19.6	6.2	12.5	-0.99	16.3
41.8	36.3	47.3	4.9	-2.9	12.2	1.2	-1.8	3.7
18.5	14.7	21.4	-4.1	-13.2	2.3	8.2	12.6	5.5
66.9	57.3	76.4	-17.0	-24.	-10.5	33.4	37.4	30.4
		Inte	er-distric	t Level				
20.8	8.7	27.6	3.5	-17.2	13.3	13.4	8.54	15.1
50.4	42.3	60.2	9.1	1.1	17.8	16.3	17.1	15.5
30.7	22.5	38.1	-5.8	-17.0	3.21	-8.6	-7.8	-9.1
38.6	29.3	48.3	-7.9	-14.6	-1.8	19.0	20.4	17.9
		Int	er-state	Level				
5.8	-3.7	13.8	-3.4	-15.8	5.4	54.5	67.6	47.1
19.7	2.3	49.2	20.1	23.8	15.8	77.5	90.9	60.8
10.4	-0.4	24.7	-9.6	-16.4	-2.5	12.0	17.2	7.3
15.9	5.9	28.2	6.06	0.87	11.2	24.9	28.3	21.7
		Total I	nternal I	Migratior	ı			
8.8	0.3	12.8	-0.03	-18.7	7.61	15.3	7.7	17.7
43.2	36.0	51.8	6.58	-1.03	14.2	22.8	27.6	18.3
21.0	14.0	27.2	-5.63	-15.2	1.9	3.0	6.4	0.7
38.6	29.2	48.7	-7.3	-14.2	-0.78	24.2	26.8	22.2
	6.2 41.8 18.5 66.9 20.8 50.4 30.7 38.6 5.8 19.7 10.4 15.9 8.8 43.2 21.0	Total         Males           6.2         -1.5           41.8         36.3           18.5         14.7           66.9         57.3           20.8         8.7           50.4         42.3           30.7         22.5           38.6         29.3           5.8         -3.7           19.7         2.3           10.4         -0.4           15.9         5.9           8.8         0.3           43.2         36.0           21.0         14.0	Intr           6.2         -1.5         9.4           41.8         36.3         47.3           18.5         14.7         21.4           66.9         57.3         76.4           Inte           20.8         8.7         27.6           50.4         42.3         60.2           30.7         22.5         38.1           38.6         29.3         48.3           Inte           5.8         -3.7         13.8           19.7         2.3         49.2           10.4         -0.4         24.7           15.9         5.9         28.2           Total I           8.8         0.3         12.8           43.2         36.0         51.8           21.0         14.0         27.2	1971-81         Total         Males         Females         Total           6.2         -1.5         9.4         -0.73           41.8         36.3         47.3         4.9           18.5         14.7         21.4         -4.1           66.9         57.3         76.4         -17.0           Inter-distric           20.8         8.7         27.6         3.5           50.4         42.3         60.2         9.1           30.7         22.5         38.1         -5.8           38.6         29.3         48.3         -7.9           Inter-state           5.8         -3.7         13.8         -3.4           19.7         2.3         49.2         20.1           10.4         -0.4         24.7         -9.6           15.9         5.9         28.2         6.06           Total Internal I           8.8         0.3         12.8         -0.03           43.2         36.0         51.8         6.58           21.0         14.0         27.2         -5.63	1971-81         1981-9           Total         Males         Females         Total         Males           Intra-district         Level           6.2         -1.5         9.4         -0.73         -19.6           41.8         36.3         47.3         4.9         -2.9           18.5         14.7         21.4         -4.1         -13.2           66.9         57.3         76.4         -17.0         -24.           Inter-district         Level         20.8         8.7         27.6         3.5         -17.2           50.4         42.3         60.2         9.1         1.1           30.7         22.5         38.1         -5.8         -17.0           38.6         29.3         48.3         -7.9         -14.6           Inter-state         Level         10.4         -0.4         24.7         -9.6         -15.8           19.7         2.3         49.2         20.1         23.8         10.4         -0.4         24.7         -9.6         -16.4           15.9         5.9         28.2         6.06         0.87         16.4           15.9         5.9         28.2         6.06	1971-81         1981-91           Total         Males         Females         Total         Males         Females           6.2         -1.5         9.4         -0.73         -19.6         6.2           41.8         36.3         47.3         4.9         -2.9         12.2           18.5         14.7         21.4         -4.1         -13.2         2.3           66.9         57.3         76.4         -17.0         -24.         -10.5           Inter-district Level           20.8         8.7         27.6         3.5         -17.2         13.3           50.4         42.3         60.2         9.1         1.1         17.8           30.7         22.5         38.1         -5.8         -17.0         3.21           38.6         29.3         48.3         -7.9         -14.6         -1.8           Inter-state Level           5.8         -3.7         13.8         -3.4         -15.8         5.4           19.7         2.3         49.2         20.1         23.8         15.8           10.4         -0.4         24.7         -9.6         -16.4         -2.5 <th< td=""><td>1971-811981-91TotalMalesFemalesTotalMalesFemalesTotal6.2-1.59.4-0.73-19.66.212.541.836.347.34.9-2.912.21.218.514.721.4-4.1-13.22.38.266.957.376.4-17.0-2410.533.420.88.727.63.5-17.213.313.450.442.360.29.11.117.816.330.722.538.1-5.8-17.03.21-8.638.629.348.3-7.9-14.6-1.819.0Inter-state Level5.8-3.713.8-3.4-15.85.454.519.72.349.220.123.815.877.510.4-0.424.7-9.6-16.4-2.512.015.95.928.26.060.8711.224.98.80.312.8-0.03-18.77.6115.343.236.051.86.58-1.0314.222.821.014.027.2-5.63-15.21.93.0</td><td>1971-811981-911991-0TotalMalesFemalesTotalMalesFemalesTotalMalesIntra-district Level<math>6.2</math><math>-1.5</math><math>9.4</math><math>-0.73</math><math>-19.6</math><math>6.2</math><math>12.5</math><math>-0.99</math><math>41.8</math><math>36.3</math><math>47.3</math><math>4.9</math><math>-2.9</math><math>12.2</math><math>1.2</math><math>-1.8</math><math>18.5</math><math>14.7</math><math>21.4</math><math>-4.1</math><math>-13.2</math><math>2.3</math><math>8.2</math><math>12.6</math><math>66.9</math><math>57.3</math><math>76.4</math><math>-17.0</math><math>-24.</math><math>-10.5</math><math>33.4</math><math>37.4</math>20.8<math>8.7</math><math>27.6</math><math>3.5</math><math>-17.2</math><math>13.3</math><math>13.4</math><math>8.54</math><math>50.4</math><math>42.3</math><math>60.2</math><math>9.1</math><math>1.1</math><math>17.8</math><math>16.3</math><math>17.1</math><math>30.7</math><math>22.5</math><math>38.1</math><math>-5.8</math><math>-17.0</math><math>3.21</math><math>-8.6</math><math>-7.8</math><math>38.6</math><math>29.3</math><math>48.3</math><math>-7.9</math><math>-14.6</math><math>-1.8</math><math>19.0</math><math>20.4</math>Inter-state Level5.8<math>-3.7</math><math>13.8</math><math>-3.4</math><math>-15.8</math><math>5.4</math><math>54.5</math><math>67.6</math><math>19.7</math><math>2.3</math><math>49.2</math><math>20.1</math><math>23.8</math><math>15.8</math><math>77.5</math><math>90.9</math><math>10.4</math><math>-0.4</math><math>24.7</math><math>-9.6</math><math>-16.4</math><math>-2.5</math><math>12.0</math><math>17.2</math><math>15.9</math><math>5.9</math><math>28.2</math><math>6.06</math><math>0.87</math><math>11.2</math><math>24.9</math><math>28.3</math>Advanted by the state of the state</td></th<>	1971-811981-91TotalMalesFemalesTotalMalesFemalesTotal6.2-1.59.4-0.73-19.66.212.541.836.347.34.9-2.912.21.218.514.721.4-4.1-13.22.38.266.957.376.4-17.0-2410.533.420.88.727.63.5-17.213.313.450.442.360.29.11.117.816.330.722.538.1-5.8-17.03.21-8.638.629.348.3-7.9-14.6-1.819.0Inter-state Level5.8-3.713.8-3.4-15.85.454.519.72.349.220.123.815.877.510.4-0.424.7-9.6-16.4-2.512.015.95.928.26.060.8711.224.98.80.312.8-0.03-18.77.6115.343.236.051.86.58-1.0314.222.821.014.027.2-5.63-15.21.93.0	1971-811981-911991-0TotalMalesFemalesTotalMalesFemalesTotalMalesIntra-district Level $6.2$ $-1.5$ $9.4$ $-0.73$ $-19.6$ $6.2$ $12.5$ $-0.99$ $41.8$ $36.3$ $47.3$ $4.9$ $-2.9$ $12.2$ $1.2$ $-1.8$ $18.5$ $14.7$ $21.4$ $-4.1$ $-13.2$ $2.3$ $8.2$ $12.6$ $66.9$ $57.3$ $76.4$ $-17.0$ $-24.$ $-10.5$ $33.4$ $37.4$ 20.8 $8.7$ $27.6$ $3.5$ $-17.2$ $13.3$ $13.4$ $8.54$ $50.4$ $42.3$ $60.2$ $9.1$ $1.1$ $17.8$ $16.3$ $17.1$ $30.7$ $22.5$ $38.1$ $-5.8$ $-17.0$ $3.21$ $-8.6$ $-7.8$ $38.6$ $29.3$ $48.3$ $-7.9$ $-14.6$ $-1.8$ $19.0$ $20.4$ Inter-state Level5.8 $-3.7$ $13.8$ $-3.4$ $-15.8$ $5.4$ $54.5$ $67.6$ $19.7$ $2.3$ $49.2$ $20.1$ $23.8$ $15.8$ $77.5$ $90.9$ $10.4$ $-0.4$ $24.7$ $-9.6$ $-16.4$ $-2.5$ $12.0$ $17.2$ $15.9$ $5.9$ $28.2$ $6.06$ $0.87$ $11.2$ $24.9$ $28.3$ Advanted by the state of the state

Table 5: Intercensal growth rates of migrants by sex and distance categories, India 1971-2001

Interestant Creveth Dat

Source: Calculated from Census of India, Migration Tables.

# 5. Econometric Framework

In the present context, to study the factors affecting the internal migration process in India, the single equation regression model is used. The theoretical background for the empirical model has been borrowed from the "Gravity Model" of migration (which includes the demographic variables viz., population size in both destinations and places of origin, and the distance between the places). At the same time, a vector of socio-economic variables is included in the same model to test the applicability of another important theory of internal migration (i.e., the Harris-Todaro model).

The origin of most empirical research of this type of "Gravity Model" of migration, in which the gross flow of migrants are assumed to be directly proportional to origin and destination population and inversely proportional to the distance between origin and destination, in addition to being dependent on a number of other origin and destination variables is:

$$M_{ij} = P_i^a P_j^b D_{ij}^g f(X_i, X_j)$$
<sup>(1)</sup>

 $\{i, j = 1, 2, \dots, n\}$ 

Where M<sub>ij</sub> is the gross flow of migrants from region i to region j, P<sub>i</sub> is the population of region i, and P<sub>j</sub> is the population of region j. D<sub>ij</sub> is the physical distance between region i and region j. And X<sub>i</sub>, X<sub>j</sub> are the vectors of origin and destination variables, respectively. Estimation of equation (1) is generally done in the double logarithmic form to get the coefficients as elasticities. In this model, the explanatory variables in the migration function typically include the urban expected wage rates (outcome of the product of urban fixed wage rate and probability of getting an urban job), the rural (agricultural) wage rate, income (Per Capita Net State Domestic Product) levels of both origin and destination places, unemployment or employment rates, education or literacy rates, and urbanisation and population levels, as well as the physical distance between regions of origin and destination.

Although there is a consensus that the types of economic variables affect migration, there is no uniformity concerning the manner in which demographic variables are entered into the migration function. A variety of dependent variables are used even in the ordinary least square (OLS) estimation; for example,  $M_{ij}/P_i$ ,  $M_{ij}/(P_i.P_j)$ ,  $M_{ij}$ . The normalisation most frequently adopted in the actual estimation is to divide migration flows by the origin population. Strict adherence to the gravity model would require the inclusion of both origin and destination population on the right of the equation. The model suggested by Levy and Wadycky (1972) is:

$$\left(\frac{M_{ij}}{P_i}\right) = P_i^{a-1} P_j^b D_{ij}^g f(X_i, X_j)$$
<sup>(2)</sup>

However, population variables included on the right-hand side often tend to be highly correlated with other explanatory variables, and the methods of their inclusion vary with the problems encountered while using different data sets. Other authors have omitted both population variables from the right-hand side or imposed a "constant return to scale" property with respect to origin and destination populations (Mundlak, 1979) - the rationale being that doubling the origin and destination

population should double the flow of migration, with factors remaining constant. Imposing homogeneity  $(a = 1-\beta)$  in equation (1) produces

$$\left(\frac{M_{ij}}{P_i}\right) = \left(\frac{P_j}{P_i}\right)^b D_{ij}^g f(X_i, X_j)$$
(3)

Similarly, the normalisation  $(M_j/P_i.P_j)$  was used by Greenwood (1971) which imposes the restriction of  $\beta=1$  (in addition to a=1) if  $P_i$ ,  $P_j$  are left out of the right hand side.

$$\left(\frac{M_{ij}}{P_i P_j}\right) = D_{ij}^g f(X_i, X_j)$$
(4)

In the present study, the dependent variable for all the migration equations (inter-state and intra-state) is the gross male migration flow divided by the relevant (rural) origin population. It is because of the fact that most of the females migrate because of non-economic reasons (i.e. marriage). The explanatory variables included in the models are not the same, because of the unavailability of data at the intra-state level. The equation 5 and 6 represent the ordinary least square (OLS) estimation model for inter-state and intra-state migrations, respectively. A detailed description of the all the explanatory variables are given in appendix-A.

$$\ln\left(\frac{M_{ij}}{P_i}\right) = \boldsymbol{b}_0 + \boldsymbol{b}_1 \ln\left(\frac{p_j}{p_i}\right) + \boldsymbol{b}_2 \ln D_{ij} + \boldsymbol{b}_3 \ln U_i + \boldsymbol{b}_4 \ln W_e + \boldsymbol{b}_5 \ln W_i + \boldsymbol{b}_6 \ln Y_i + \boldsymbol{b}_7 \ln Y_j + \boldsymbol{b}_8 \ln L_i + \boldsymbol{b}_9 \ln Lit_i + \boldsymbol{b}_{10} \ln GFCF_i + \boldsymbol{b}_{11} \ln Wat_j + \boldsymbol{b}_{12}LD + \boldsymbol{e}_1$$
(5)

$$\ln\left(\frac{M_{ij}}{P_i}\right) = \boldsymbol{b}_0 + \boldsymbol{b}_1 \ln\left(\frac{p_j}{p_i}\right) + \boldsymbol{b}_2 \ln D_{ij} + \boldsymbol{b}_3 \ln U_i + \boldsymbol{b}_4 \ln W_e + \boldsymbol{b}_5 \ln W_i + \boldsymbol{b}_6 \ln Y_i + \boldsymbol{b}_7 \ln Y_j + \boldsymbol{b}_8 \ln L_i + \boldsymbol{b}_9 \ln Lit_i + \boldsymbol{b}_{10} \ln GFCF_i + \boldsymbol{b}_{11} \ln Wat_j + \boldsymbol{b}_{12}LD + \boldsymbol{e}_2$$
(6)

Estimation of determinants of migration using single equation methods has been the practice in almost all previous works and hence is given prime importance in the present study. In the present study, an attempt has also been made to study the simultaneous relationship between migration rate and rural wage, using a two-stage least square (2SLS) method. The existence of simultaneous relationship between the rural wage and internal migration was established by Harris-Todaro (1970), where they have assumed that in the urban sector, there exists an institutional fixed wage, which is above the minimum subsistence level. The internal migration is driven by the higher expected wages in urban sector, and the equilibrium rural wage and migration flows are determined simultaneously in the economy. The Harris-Todaro condition for long run equilibrium is:

$$W_e - W_i = 0 \tag{7}$$

The equation 8 and 9 are the structural equations for both migration and rural wage functions estimated by the Two-Stage Least Square (TSLS) method at inter-state and intra-state levels. A detailed description of the all the explanatory variables are given in appendix-A.

$$\ln\left(\frac{M_{ij}}{P_i}\right) = \boldsymbol{b}_0 + \boldsymbol{b}_1 \ln\left(\frac{p_j}{p_i}\right) + \boldsymbol{b}_2 \ln D_{ij} + \boldsymbol{b}_3 \ln U_i + \boldsymbol{b}_4 \ln W_e + \boldsymbol{b}_5 \ln W_i + \boldsymbol{b}_6 \ln Y_i + \boldsymbol{b}_7 \ln Y_j + \boldsymbol{b}_8 \ln L_i + \boldsymbol{b}_9 \ln Lit_i + \boldsymbol{b}_{10} \ln GFCF_i + \boldsymbol{b}_{11} \ln Wat_j + \boldsymbol{b}_{12}LD + \boldsymbol{e}_1$$
(8A)

$$\ln W_i = \boldsymbol{b}_0 + \boldsymbol{b}_1 \ln U_i + \boldsymbol{b}_2 \ln Y_i + \boldsymbol{b}_3 \ln L_i + \boldsymbol{b}_4 \ln \left(\frac{M_{ij}}{P_i}\right) + \boldsymbol{b}_5 \ln Lit_i + \boldsymbol{b}_6 \ln GFCF_i + \boldsymbol{b}_7 \ln Emp_i + \boldsymbol{e}_2$$
(8B)

$$\ln\left(\frac{M_{ij}}{P_i}\right) = \boldsymbol{b}_0 + \boldsymbol{b}_1 \ln\left(\frac{p_j}{p_i}\right) + \boldsymbol{b}_2 D + \boldsymbol{b}_3 \ln U_i + \boldsymbol{b}_4 \ln W_e + \boldsymbol{b}_5 \ln W_i + \boldsymbol{b}_6 \ln Emp_i + \boldsymbol{b}_7 \ln L_i + \boldsymbol{b}_8 \ln Lit_i + \boldsymbol{e}_1$$
(9A)

$$\ln W_i = \boldsymbol{b}_0 + \boldsymbol{b}_1 \ln U_i + \boldsymbol{b}_2 \ln L_i + \boldsymbol{b}_3 \ln \left(\frac{M_{ij}}{P_i}\right) + \boldsymbol{b}_4 \ln Lit_i + \boldsymbol{b}_5 \ln GFCF_i + \boldsymbol{b}_6 \ln Emp_i + \boldsymbol{e}_2$$
(9A)

## 6. Rural to Urban Migration: Empirical Results:

The migration function estimated on the basis of aggregated data may not be preferable to base our expectations about the results on a single hypothesised response. Because of the fact that people who belongs to different economic sections will respond differently to changes in economic variables. Again individuals in the same migration stream may have widely different reasons for moving. Accordingly, an attempt is made here to offer, wherever necessary, competing explanations of a particular result. Since all the variables are in the form of logarithms, the parameter estimates represent elasticities. The T-statistics are presented in parentheses along with the coefficient values. A single, double and triple star implies the statistical level of significance at 10%, 5% and 1%, respectively.

# 6.1. Inter-state Rural to Urban Migration

Both Ordinary Least Square (OLS) and Two-Stage Least Square (TSLS) results of the less than one year and one to four years durations of inter-state rural-urban migration are presented in Table 6 & 7 to facilitate comparisons. The OLS results are presented in Column 2 in both the tables, where as the TSLS results are given in Column 3 & 4 respectively.

The OLS results presented here are the robust result which was corrected for both heteroscedasticity and possible serial correlation problems. The mean of variance inflation factors figures are given in the tables and they indicate that the degree of multicollinearity is very low, though

it does not affect the estimated coefficients. The  $R^2$  (the measure of goodness of fit) for both the equations are quite reasonable along with high significance levels of F-statistics.

The estimated results suggest that in both durations of migration, the gravity variables (population size and distance) are statistically (at 1% level) as well as economically significant, with positive and negative signs (as expected), and absolute values of the coefficients are 0.58 & 0.608 (population elasticities of migration) and -1.3 & -1.48 (distance elasticities of migration) respectively. The origin population size acts as a pushing factor where as the size of urban population is an indication for the rate of urbanisation (as a pulling factor). Both origin and destination population have positive roles to play in the migration process, which is evident from the empirical results. The distance is the proxy for all migration costs (including the psychic cost) and has played a deteriorating role in the internal migration process in India.

Variables	OLS	25	LS
Variables	M <sub>ij</sub> /P <sub>i</sub>	M <sub>ij</sub> /P <sub>i</sub>	Wi
Constant	-12.23(-5.01)***	-18.78(-5.0)***	0.33(0.72)
P <sub>j</sub> /P <sub>i</sub>	0.58(16.29)***	0.65(15.14)***	
D <sub>ij</sub>	-1.3(-14.44)***	-1.41(-12.07)***	
Ui	0.49(4.83)***	0.38(3.08)***	0.032(1.80)*
We	0.009(0.15)	-0.027(-0.37)	
Wi	0.13(0.63)	4.14(3.46)***	
Yi	-0.55(-2.44)***	-1.69(-3.87)***	0.27(6.91)***
Yj	1.35(10.05)***	1.42(7.92)***	
Li	-0.07(-0.78)	0.55(2.51)**	-0.16(-9.90)***
Lit <sub>i</sub>	-0.7(-1.13)	-0.63(-0.72)	0.18(1.36)
GFCFi	0.24(5.47)***	0.36(5.99)***	-0.055 (-5.35)**
Watj	0.32(1.73)*	0.21(0.87)	
LD	0.71(4.34)***	0.84(3.94)***	
M <sub>ij</sub> /P <sub>i</sub>			-0.01(-1.59)
Empi			-0.03(-3.4)***
Pi			0.062(5.38)***
R <sup>2</sup>	0.637	0.41	0.34
F-Statistics	76.37***	48.97***	34,49***
Root MSE	1.27	1.61	0.23
Mean VIF	1.64		
N	543	543	543

Table 6: Interstate Rural-Urban Migration (Duration of less than 1 year)

Note: The T-statistics are presented in parentheses and \*, \*\*, and \*\*\* implies the statistical level of significance at 10%, 5% and 1%, respectively.

The origin (rural) unemployment has the positive coefficients (0.49 and 0.43) as expected and are highly significant (at 1% level), suggesting that the origin unemployment is one of the major pushing factors responsible for rural-urban migration. But the most crucial finding of the study is the insignificant expected wage, the most important economic variable as explained in the famous 'Harris-

Todaro' model. The rapid expansion of the urban informal sector in India could be responsible for this, which in itself attracts the workers. There hardly exists any labour mobility from urban informal to the urban formal sectors which is shown by the fact that about 93% of the total work force is engaged in informal sector in India (Sengupta, 2007). The coefficient of rural agricultural real wage has a limited role to play in the inter-state migration, since it is statistically insignificant in all the equations. It may be due to the fact that the migration is not just by the very poor during times of crisis for survival and coping but has increasingly become an accumulative option for both poor and non-poor (Deshingkar and Start, 2003). The coefficients of both origin and destination states' per capita net state domestic product are highly statistically (at 1% level) as well as economically significant (having expected negative and positive signs, respectively). This finding is similar to Greenwood study (1997) which stated that the per capita gross domestic product (GDP) is the most representative macro-economic variable responsible for migration of people as high economic prosperity means more activities, services and opportunities for people living in that area. The results suggest that origin income elasticities of migration are less than one (-0.55 and -0.56) where as the destination income elasticities of migration are greater than one (1.35 in both the equations). This suggests that the rural to urban migration in India is basically to the relatively developed states.

Variables	OLS	25	LS
variables	M <sub>ij</sub> /P <sub>i</sub>	M <sub>ij</sub> /P <sub>i</sub>	Wi
Constant	-8.89(-3.43)***	-22.93(-4.43)***	0.94(2.26)**
Pj/Pi	0.608(18.29)***	0.7(14.45)***	
D <sub>ij</sub>	-1.48(-16.54)***	-1.64(-11.46)***	
Ui	0.43(4.02)***	0.079(0.48)	0.06(3.81)***
We	-0.029(-0.57)	-0.068(-0.76)	
Wi	0.34(1.52)	6.82(4.39)***	
Yi	-0.56(-2.44)***	-2.56(4.40)***	0.29(8.03)***
Yj	1.35(9.53)***	1.45(6.53)***	
Li	-0.18(-1.74)*	0.82(2.96)***	-0.16(-10.65)***
Lit <sub>i</sub>	-1.06(-1.62)	0.24(0.23)	0.002(0.02)
GFCFi	0.23(5.41)***	0.36(5.57)***	-0.044(-4.99)**
Watj	0.49(2.42)***	0.45(1.57)	
LD	0.64(3.54)***	0.74(2.80)***	
M <sub>ij</sub> /P <sub>i</sub>			-0.007(-1.49)
Empi			-0.038(-3.62)**
Pi			0.054(5.49)***
R <sup>2</sup>	0.639	0.179	0.34
F-Statistics	91.3***	41.18***	40.17***
Root MSE	1.40	2.11	0.23
Mean VIF	1.59		
N	624	624	624

 Table 7: Interstate Rural-Urban Migration (Duration of 1 to 4 years)

Note: The T-statistics are presented in parentheses and \*, \*\*, and \*\*\* implies the statistical level of significance at 10%, 5% and 1% respectively.

The coefficient of the average size of operational holdings is significant only in 1 to 4year duration case with a negative sign (-0.18), implying the fact the people with small operational holdings prefer to migrate to the urban sector and either settles down there permanently or send remittance regularly back home. The coefficients of Gross Fixed Capital Formation (GFCF) in agriculture are highly significant, with positive signs (0.24 and 0.23), implying that the people from agriculturally advanced regions migrate to urban sectors. The reasons could be either that they can bear the migration costs and have a greater propensity to migrate for better opportunities or to carry out their businesses.

The coefficients of rural literacy rate are not significant in all the regression equations. This prompts us to think again about the urban informal sector in which a large proportion of workers are unskilled and illiterate. The coefficient of the water facilities at the destination is positive and significant in both the cases (0.32 and 0.49), indicating urban sanitation facility is one the factors responsible for rural-urban migration in India. The coefficients language dummies are positive and highly significant (0.71 and 0.64) in both equations suggesting that language plays an important role in inter-state migration in India. Out of the total migrants, 71% of migrants in the less than 1 duration and 64% of migrants in the 1 to 4 years of duration scenarios prefer to migrate to states with similar language.

Two-Stage Least Square estimates are presented above to facilitate the comparison with the OLS results. After satisfying both the necessary and sufficient identification conditions (i.e., the rank and order conditions for identification) for all equations, it is clear form both Table 6 and 7 that the simultaneous relationship between migration rate and the average rural wage is not established even though the 'Hausman Test' suggests the possibility of the existence of simultaneity between migration rate and rural wage. The coefficients of migration rates are insignificant in both the rural wage equations even if rural wage rates are highly significant in both the cases. This 2SLS results again put a question mark on the applicability of 'Harris-Todaro' model in the case of a developing country like India

#### 6.2. Intrastate Rural to Urban Migration:

The estimated results in intra-state rural-urban migration of both ordinary least square (OLS) and twostage least square (2SLS) for less than one year and one to four years durations are presented in Table 8 & 9 respectively. Again, the OLS results presented here are robust result with very low degree of multicollinearity, considerable  $R^2$  and highly significant F-statistics.

In the intra-state case too, the "Gravity Model" is well established in India. The coefficients of the destination and origin population ratio are positive and highly significant, which supports the theoretical argument that the internal migration is directly proportional to the population size of both origin and destination places. And the coefficients of distance dummies are negative (-0.35 & -0.408) in both the less than 1 year and 1 to 4 years of duration cases, as expected and highly significant. This indicates the fact that distance factor has a deteriorating impact on population mobility.

The coefficient of Harris-Todarian expected wage rate is highly significant in less than 1 year duration of migration where as insignificant in 1 to 4 year duration of migration cases. This result, in fact, is very interesting since most of the short distance and duration migrants are seasonal migrants and commuters, often guided by the higher expected wages in the destination places. But as the duration of migration increases, there are a host of factors responsible for rural-urban migration where the role of

expected wage is almost restricted. The coefficient of origin unemployment is statistically significant having the expected negative signs (-0.22 and -0.31) in both less than one year and one to four years of duration cases, which suggests the fact that the origin unemployment is one of the major factors for intra-state (short distance) rural-urban migration. But the most interesting result that appears in within state migration is that the coefficient of the destination employment is statistically significant with the negative signs (-5.08 and -1.34). This implies the fact that the short distance migration in India is discouraged by the destination employment because the urban informal sector attracts both unskilled and semiskilled workers from the rural areas. Both the self employed and wage workers face competition in the informal job market and it is very difficult for them to get a livelihood if the destination places are already crowded.

Maniahlaa	OLS	2SL	S	
Variables	M <sub>ij</sub> /P <sub>i</sub>	M <sub>ij</sub> /P <sub>i</sub>	Wi	
Constant	10.11(1.61)	9.23 (1.28)	4.48(6.10)***	
P <sub>j</sub> /P <sub>i</sub>	0.16(1.93)*	0.14(1.71)*		
D	-0.35(-4.17)***	-0.35(-4.17)***		
Ui	-0.22(-1.80)*	-0.21(-1.72)*	0.044( 0.74)	
We	0.98(4.25)***	1.02( 3.82)***		
Wi	-0.01(-0.10)	-0.095(-0.39)		
Empj	-5.08(-3.25)***	-4.82(-2.66)***		
Li	0.08(1.23)	0.081( 0.95)	0.081(0.95)	
Lit <sub>i</sub>	0.84(7.58)***	0.86( 6.14)***	0.18(2.01)**	
M <sub>ij</sub> /P <sub>i</sub>			-0.037(-0.53)	
R <sup>2</sup>	0.301	0.30	0.05	
F-Statistics	46.75***	21.58***	4.87***	
Root MSE	0.86	0.86	0.39	
Mean VIF	1.88			
Ν	410	410	410	

Table 8: Rural-Urban Migration (Duration of less than 1 Year)

Note: The T-statistics are presented in parentheses and \*, \*\*, and \*\*\* implies the statistical level of significance at 10%, 5% and 1% respectively.

The coefficients of the average size of operational holdings are insignificant in both equations suggesting that land holding does not have any impact on the intra-state migration flows. The coefficients of rural literacy are highly significant, with positive signs (0.84 and 0.86) in all the regression equations, indicating that literacy plays a crucial role in intra-state migration. This could be due to the reason that among the class of migrants, the educated people have a greater chance of migrating, for purpose of either higher education or in search of better job opportunities.

Similarly, the Two-Stage Least Square estimates presented above in Table 8 and 9 fail to establish simultaneous relationship between migration rate and the average rural wage as the coefficient of rural wage is insignificant in both the migration equations. The coefficient of migration rate is not significant in the rural wage equation in both durations even though 'Hausman Test'

suggested the possibility of the existence of simultaneity between migration and rural wage. The OLS results itself provide consistent estimator for the parameters estimated in the intra-state migration.

Mantahlar	OLS	2SL	S	
Variables	M <sub>ij</sub> /P <sub>i</sub>	M <sub>ij</sub> /P <sub>i</sub>	Wi	
Constant	-1.9(-0.32)	-3.22(-0.45)	4.48(6.2)***	
P <sub>j</sub> /P <sub>i</sub>	0.11(1.27)	0.08(1.0)		
D	-0.408(-4.86)***	-0.408(-4.85)***		
Ui	-0.31(-2.56)***	299(-2.42)***	0.04(0.48)	
We	0.28(1.28)	0.34(1.3)		
Wi	0.05(0.53)	-0.06(-0.28)		
Empj	-1.34(-0.9)	-0.95(-0.54)		
Li	0.064(0.95)	0.061(0.73)	-0.03(-0.80)	
Lit <sub>i</sub>	0.86(7.77)***	0.899(6.48)***	0.19(1.85)*	
M <sub>ij</sub> /P <sub>i</sub>			-0.04(-0.55)	
Pi			-0.03(-0.8)	
R <sup>2</sup>	0.277	0.275	0.042	
F-Statistics	46.16***	19.15***	4.84***	
Root MSE	0.85	0.85	0.39	
Mean VIF	1.88			
Ν	410	410	410	

Table 9: Rural-Urban Migration (Duration of 1 to 4 Years)

Note: The T-statistics are presented in parentheses and \*, \*\*, and \*\*\* implies the statistical level of significance at 10%, 5% and 1% respectively.

# 7. Summary and Conclusion

The rural to rural migration has been the most important migration flow that indicates a steadily declining proportion, while the proportions of other categories have increased over the period. In the same way, the proportion of short distance migrants has decreased while the proportions of medium and long distance migrants have increased, suggesting that the long distance movements are more urban-oriented than short distance movements. The short distance migration is main migration pattern among Indian females while long distance migration is often undertaken by the males. Marriage is the reason for a large proportion female migration where as in the case of males, the most important reasons are employment, business etc. Again, it is evident from the Indian Census that there has been a significant increase in migration to urban areas both among males and females during 1991-2001. The rural to urban migration may be due to the rapid growth of urban informal sectors in the recent years, which comprises about 93 per cent of the total employment in the economy. The movements from urban to urban areas are also increasing considerably. This may be due to globalisation and the quick expansion of the service sector. From the current trends and patterns of internal migration in India, it can be anticipated that long distance rural to urban and urban to urban streams will be the dominant migration streams in the future.

In this context, the study analysed the determinants of migration to provide a sound understanding of the factors responsible for rural to urban migration. The estimated results from OLS method suggests that in both inter-state and intra-state levels and durations (less than 1 year and one to four years) of migration, the gravity variables (population size and distance between the places) are highly significant, with the expected signs while the Harris-Todarian variable dealing with the urban expected wages, has a limited role to play in the internal migration process in India. At the same time, the possibility of the existence of simultaneous relationship between the migration rates and rural (agricultural) wage rate is rejected in both the categories of migration in India. But there exists other push and pull factors (like unemployment, level of income, average size of operational holdings, urban water and sanitation facilities, similar language etc.) which operating mutually both at origin and destination places, are the main determinants for internal migration. The role of the third sector (urban informal sector) in the internal migration process is not explicitly analysed here, which is one of the limitations of the present macro-level study. The main reason for this is the use of the group-level data of Indian Census, which fails to provide information about a particular migrant, whether he/she is migrating to urban informal sector or formal sector. However, these limitations of Census data can be tackled if they include a few more questions in addition to the question of the place of last residence. It would be better if they ask whether they are staying at the same place or have ever changed their place of residence; if yes since how long? With regard to the employment question, it is worthwhile to include an additional question which can explicitly define the informal sector employment since it constitutes the major share in employment in Indian labour market. In order to get information regarding seasonal migration, a question can be asked whether the particular respondent moves elsewhere in a particular year for the purpose of employment during the agricultural off seasons. Information about commuting can easily be collected by asking whether his place of work is same as the place of residence. Nevertheless, a detailed and in-depth study of migration behaviour (at the micro-level) and the labour market dynamics in India is urgently required. The forthcoming National Sample Survey (NSS 64<sup>th</sup> Round) data pertaining to migration details and in-depth micro survey related to new directions and issues may be useful to examine the complexity arising out of the context of migration patterns in India.

# Notes

The urban expected wage  $W_e = W_j$  (Emp<sub>j</sub>/Total urban labour force) is equated to the rural wage  $W_i$  (here in the origin wage rates) at the long run equilibrium where the migration flow from rural to urban sector comes to a halt. Therefore, the rural wage rates and migration rates in short-run are determined simultaneously (Harris J. & M. Todaro, 1970).

Hausman test is performed to find out whether a particular variable is exogenous or endogenous in the model. A variable x is exogenous if the conditional distribution of y given x does not change with modifications of the process generating x. And a *strictly exogenous* variable is one that is independent of all contemporaneous, future and past errors (Wooldridge, 2006; and Brooks, 2008)

# Reference

- Agesa, Richard U (2001). Migration and the Urban to Rural Earnings Difference: A Sample Selection Approach, *Economic Development and Cultural Change*, Vol. 49, No. 4, pp 847-865.
- Andrienko, Yuri, Guriev, Sergei (2003). Determinants of Interregional Mobility in Russia: Evidence from Panel Data, *Economics of Transition*, Vol. 12, pp 1-2.
- Banerjee, B (1998). Migration Motivation, Family links, and Job Search Methods of Rural-to-Urban Migration in India. In R Bilsborrow (ed), *Migration, Urbanization and Development: New Directions and Issues.* New York: United Nations Population Fund and Kluwer Academic Publishers, pp 187-219.
- Barkley, Andrew P (1990). The Determinants of the Migration of Labor out of Agriculture in the United States, 1940-85, *American Journal of Agricultural Economics*, Vol. 72, No. 3, pp 567-573.
- Barro, Robert J and Xavier Sala-i-Martin (1991). Convergence Across States and Regions, Brookings Papers on Economic Activity, no. 1, pp 107-182.
- Bhagat, R (2005). Conceptual Issues in Measurement of Internal Migration in India, Paper Presented in XXV<sup>th</sup> IUSSP International Conference held in Tours, France, July 18-23.
- Bhattacharya, P C (1998). The Informal sector and Rural-to-Urban Migration: Some Indian Evidence, *Economic political weekly*, May 23.
- Bhattacharya, P C (2002). Urbanisation in Developing Countries, *Economic and Political Weekly*, October 12.
- Bilsborrow, R (1998). The State of the Art and Overview of the Chapters. In R Bilsborrow (ed), *Migration, Urbanization and Development: New Directions and Issues.* New York: United Nations Population Fund and Kluwer Academic Publishers, pp 1-56.
- Brooks C (2008). Introductory Econometrics for Finance, 2<sup>nd</sup> edition. Cambridge: CUP.
- Census of India, Migration tables, D-Series for 1971, 1981, 1991 & 2001.
- Deshingkar *et al* (2006). The Role of Migration and Remittances in Promoting Livelihoods in Bihar, DFID research project.
- Deshingkar, P and D Start (2003). Seasonal Migration for Livelihoods in India: Coping, Accumulation and Exclusion, working paper 220.
- Deshingkar, P (2006). Internal Migration, Poverty and Development in Asia, http://wwwasia2015conference.org.
- Deshingkar, P (2004). Understanding the Implications of Migration for Pro-poor Agricultural Growth, paper prepared for the DAC POVNET Agriculture Task Group Meeting, Helsinki, June 17–18.
- Dhar S (1980). Interstate and within-state Migration in India. An analysis of internal migration in India, Yale University Press.
- Diehl, William D (1996). Farm-Nonfarm Migration in the Southeast: A Costs-Returns Analysis, *Journal of Farm Economics*, Vol. 48, No. 1, pp 1-11.
- Greenwood M J (1971). A Regression Analysis of Migration to Urban Areas of a Less-Developed Country: The Case of India. *Journal of Regional Science*. Vol. 11, No. 2.
- ———— (1985). Human migration: Theory, models and empirical studies, Journal of Regional Science Vol. 25, pp 521–544.

- ———— (1997), "Internal Migration in Developed Countries. In Mark R Rosenzweig and Oded Stark (ed), Handbook of Population and Family Economics, Volume 1B. Amsterdam: Elsevier, pp. 647-720.
- Gupta, M (1993). Rural-urban Migration, Informal sector and Development Policies: A Theoretical Analysis, *Journal of Development Economics*, Vol. 41, No. 1, pp 137-151.
- Harris, J R and M P Todaro (1970). Migration, Unemployment and development: A two sector analysis, *American Economic Review*, Vol. 60, No.1, pp 126-38.
- Hossain, M Z (2001). Rural-Urban Migration in Bangladesh: A Micro-Level Study, Paper prepared for presentation in a Poster Session on INTERNAL MIGRATION at the Brazil IUSSP Conference during August 20-24. http://agcensus.nic.in/cendata/databasehome.aspx.
- Joshi G V and N Lobo (2003). Rural-urban migration and rural unemployment in India. New Delhi: Mohit.
- Kothari, U (2002). Migration and Chronic Poverty, Working Paper No 16, Chronic Poverty Research Centre, Institute for Development Policy and Management, University of Manchester.
- Kundu and Sarangi (2007). Migration, Employment Status and Poverty An Analysis across Urban Centres, *Economic and Political Weekly*, January 27.
- Kundu, Amitab and Gupta, Salini (1996), "Migration, Urbanization and Regional Inequality", Economic and Political weekly, December, 28.
- Larson, Donald and Yair Mundlak (1995). On the intersectorial migration of agricultural labor, Policy Research Working Paper.
- Levy, M B and Walter JWadycky (1974). Education and the decision to migrate: and econometric analysis of migration in Venezuela, *Econometrica*, Vol. 42, No. 2, pp 377-388.
- Lewis, A (1954). Economic development with unlimited supplies of labour, Manchester School of Economic and Social Studies, May.
- Lusome, R and R B Bhagat (2006). Trends and Patterns of Internal Migration in India, 1971-2001, Paper presented at the Annual Conference of Indian Association for the Study of Population (IASP) during 7-9 June, 2006, Thiruvananthapuram.
- Mincer, J (1978). Family migration decisions. Journal of Political Economy, Vol. 86, pp 749-73.
- Ministry of Agriculture (2000-01). Agricultural Wages in India, Directorate of economics and statistics. New Delhi: Govt. of India.
- Mitra, A (2003). Urban Labour Market: Networks and Physical Segmentation, Occupational Choices, Networks and Transfers: An exegesis Based on Micro Data from Delhi Slums, Chapter 3.
- Mitra, A and M Murayama (2008). Rural to Urban Migration: A District Level Analysis for India, *IDE discussion paper no.137*.
- Mundlak (1979). Agriculture; Rural development; Rural-urban migration; Economic aspects; Mathematical models; Japan. *Intersectoral factor mobility and agricultural growth*, ISBN<u>.</u>
- Pekkala, S and Hannu Tervo (2002). Unemployment and Migration: Does Moving Help?, *The Scandinavian Journal of Economics*, Vol. 104, No. 4, pp 621-639.
- Perloff, J M *et al* (1998). Migration of Seasonal Agricultural Workers, *American Journal of Agricultural Economics*, Vol. 80, No. 1, pp 154-164.

- Ranis, G and John C H Fei (1961). A Theory of Economic Development, *The American Economic Review*, Vol. 51, No. 4, pp 533-565.
- Sengupta, A (2007). Report on Condition of work and promotion of livelihoods in the unorganised sector, National Commission for Enterprises in the Unorganized Sector (2006), *NCEUS Repot* on Social Security for Unorganised Workers, presented to the Prime Minister, Government of India.
- Singh, D P (1998). Internal Migration in India: 1961-1991, Demography India, Vol. 27.
- Srivastava, R (1998), Migration and the labour market in India, *The Indian Journal of Labour Economics*, Vol. 41, No. 4.
- Srivastava, R and S K Sasikumar (2003). An Overview of Migration in India, Its Impacts and Key Issues, Refugee and Migratory Movements Research Unit, Bangladesh and DFID, UK.
- Stark, O and E Katz (1986). Labour Migration and Risk Aversion in Less Developed Countries, *Journal Of Labour Economics*, Vol. 4 No. 1, pp 134-49.
- Stark, O (1991). The Migration Of Labour", Cambridge, Basil Blackwell Publication.
- Stark, O and Bloom D E (1985). The New Economics Of Labour Migration, *American Economic Review*, Vol. 75, No. 2, pp 173-178.
- Taylor, J E and P L Martin (2001). Human Capital: Migration and Rural Population Change. In Gardner,
   B and G Rausser (ed), Hand Book Of Agricultural\_Economics, Vol. 1, Amsterdam Elsevier
   Publication, pp.457-511.
- Todaro, M P (1969). A Model of Labor Migration and Urban Unemployment in Less Developed Countries, *The American Economic Review*, Vol. 59, No. 1, pp 138-148.
- Ullah, A K M Ahsan (2004). Bright City Lights and Slum of Dhaka city: Determinants of rural-urban migration in Bangladesh, *migration letters*, Vol. 1, No. 1, pp 26-41, October.
- Waddington, H and R Sabates-Wheeler (2004). How does poverty affect migration choice, a review of the literature, Working Paper T3, Development Research Centre on Migration, Globalisation & Poverty, Sussex Centre for Migration Research, Brighton: University of Sussex.
- Wooldridge, J M (2006). Introductory Econometrics: A modern Approach. South-Western Cengage Learning, 3<sup>rd</sup> edition (India), New Delhi.

#### Websites:

http://agcensus.nic.in/cendata/databasehome.aspx http://wwwasia2015conference.org www.indiastat.com. www.mapsofindia.com/distance

# Appendix

# 1. Variables used in the analysis of inter-state migration in India

NA	Number of males whose providus place of residence was state i (rural) and whose present
M <sub>ij</sub>	Number of males whose previous place of residence was state i (rural) and whose present place of residence (2001) was state j (urban) by the duration of residence in the destination
	urban place.
Pi	Total rural population of state i
Pj	Total urban population of state j
Úi	Estimated of Rural Unemployment Rate for Youth (15 to 29 Years) in state i as per usual
	status approach.
Uj	Estimated of Urban Unemployment Rate for Youth (15 to 29 Years) in state j as per usual
	status approach.
Y <sub>i</sub>	Per Capita Net State Domestic Product (NSDP) at Constant 1999-2000 Prices for state i
Yj	Per Capita Net State Domestic Product (NSDP) at Constant 1999-2000 Prices for state j
Wi	Average Daily Wage Rates for Agricultural Occupation (Ploughing) in state i
Wj	Labour Cost on Wages/Salaries of Workers per Man-day Worked at the industrial sector in
	state j
W <sub>e</sub>	The urban expected wage is calculated by multiplying W <sub>j</sub> and Emp <sub>j</sub> and dividing the product
	by 100.
Lit i	The Rural Literacy rate of state i.
Lit <sub>j</sub>	The urban Literacy rate of state j.
GFCF	Gross fixed capital formation in agriculture (irrigation) in state i
Li	Average size of operational landholdings in state i
Empi	Rural employment rates in state i
Emp <sub>i</sub>	Urban employment rates in state j
Wati	Percentage of urban population getting water facility in state j
D <sub>ij</sub>	The physical distance between the capital cities of state i and j
LĎ	The language dummy takes value 1 for same language in origin and destination states and
	zero otherwise.

# 2. Variables used in the analysis of intrastate migration in India

M <sub>ij</sub>	Number of males whose previous place of residence was places i (rural) within or outside
	the district of enumeration in a particular state and whose present place of residence
	(2001) was place j (urban) by the duration of residence in the destination urban place.
Pi	Total rural population of district i
Pj	Total urban population of the state
Ui	Estimated of Rural Unemployment Rate (15 to 59 Years) in district i
Uj	Estimated of Urban Unemployment Rate for Youth (15 to 29 Years) in State
Wi	Average Daily Wage Rates for Agricultural Occupation (Ploughing) in district i
Wj	Labour Cost on Wages/Salaries of Workers per Man-day Worked at the industrial sector in
	state
W <sub>e</sub>	The urban expected wage is calculated by multiplying $W_j$ and $Emp_j$ and dividing the product
	by 100.
Lit i	The Rural Literacy rate of district i.
Li	Average size of operational landholdings in district i
Emp <sub>i</sub>	Urban employment rates in state
Wat	Percentage of urban population with water facility in state
D <sub>ii</sub>	Distance dummy assumes value 1 for the migrants whose previous place of residence was
Í	outside the district of enumeration and zero for within the districts of enumeration
1	

# **Recent Working Papers**

- 170 Non-Agricultural Employment for Young Women in India: Status, Opportunity and Ways Forward D Rajasekhar
- 171 Community Contribution for Enviornmental Sanitation: Myth or Reality? Veerashekharappa
- 172 Does Repayment Indicate the Success of Micro-Finance Programme? Emil Mathew
- 173 Community Participation in Rural Water Supply: An Analysis Using Household Data from North Kerala Nisha K R
- 174 Urbanisation in a Forward Looking Statpe of India: Patterns Issues and Policy G S Sastry
- 175 Contract Labour Act in India: A Pragmatic View Meenakshi Rajeev
- 176 Issues of Unaccounted for Water in the Urban Water Sector G S Sastry
- 177 Liberalisation and Efficiency of Indian Commercial Banks: A Stochastic Frontier Analysis H P Mahesh
- 178 Power Sharing in the Panchayats of Orissa Pratyusna Patnaik
- 179 Can Career-Minded Young Women Reverse Gender Discrimination? Alice W Clark and T V Sekher
- **180 People's Participation in Environmental Protection: A Case Study of Patancheru** Geetanjoy Sahu
- **181 Efficiency and Bureaucracy** Anitha V
- 182 Reproductive and Child Health Programmes in the Urban Slums of Bangalore City: A Study on Unmet Needs fro Family Welfare Services C S Veeramatha
- 183 Demographic Change and Gender Inequality: A Comparative Study of Madhya Pradesh and Karnataka C M Lakshmana
- 184 Increasing Ground Water Dependency and Declinin Water Quality in Urban Water Supply: A Comparative Analysis of Four South Indian Cities K V Raju, N Latha and S Manasi
- 185 Impact of Land Use Regulations on Suburbanisation: Evidence from India's Cities Kala Seetharam Sridhar
- 186 Socio-Economic Determinants of Women Leadeship at the Grass - Roots K C Smitha
- 187 Groundwater for Agriculural Use in India: An Institutional Perspective Sarbani Mukherjee

- 188 Comparative Study of Traditional Vs. Scientific Shrimp Farming in West Bengal: A Technical Efficiency Analysis Poulomi Bhattacharya
- 189 Urban and Service Delivery in Bangalore: Public-Private Partnership Smitha K C and Sangita S N
- 190 Social Capital in Forest Governance Regimes Sangita S N
- 191 Agriculture in Karnataka: A Historical View After the Fall of Serirangapatana R S Deshpande and Malini Tantri
- **192 Personality Traits and Administrators** Anitha V
- 193 Sustainability of Indian Agriculture: Towards an Assessment V M Rao
- 194 Emerging Development Issues of Greater Bangalore G S Sastry
- 195 Rural Infrastructure Development Fund: Need for a Track Change Meenakshi Rajeev
- 196 Emerging Ground Water Crisis in Urban Areas — A Case Study of Ward No. 39, Bangalore City K V Raju, S Manasi and N Latha
- 197 In Pursuit of India's Export earning advantage: An Assessment of IT-Enabled Services Industry Meenakshi Rajeev
- **198 A Patriarchal Link to HIV/AIDS in India** Skylab Sahu
- 199 Collective Action and Property Rights: Some Critical Issues in the Context of Karnataka K G Gayathri Devi
- 200 State, Society and Inclusive Governance: Community Forests in Andhra Pradesh, Karnataka and Orissa S N Sangita
- 201 Urban Poverty and Links with the Environment: An Exploration K G Gayathri Devi
- 202 Groundwater Over-exploitation, Costs and Adoption Measures in the Central Dry Zone of Karnataka Anantha K H and K V Raju
- 203 Changing Child Population: Growth, Trends and Levels in Karnataka C M Lakshmana
- 204 Awareness About HIV/AIDS Among Karnataka Woment: An Analysis of RCH 2002-04 Data K S Umamani
- 205 The Microfinance Promise in Financial Inclusion and Welfare of the Poor: Evidence from Karnataka, India Naveen K Shetty
- 206 Structure of Central Himalayan Forests Under Different Management Regimes: An Empirical Study Sunil Nautiyal

- 207 Poverty and Natural Resources: Measuring the Links (Some Issues in the Context of Karnataka) K G Gayathri Devi
- 208 Federalism and Decentralisation in India: Andhra Pradesh and Tamil Nadu V Anil Kumar
- 209 Capital, 'Development' and Canal Irrigation in Colonial India Patric McGinn
- 210 Gender, Ecology and Development in Karnataka: Situation and Tasks Ahead K G Gayathri Devi
- 211 Greenhouse Gases Emission and Potential Corbon Sequestration: A Case Study of Semi-Arid Area in South India Lenin Babu and K V Raju
- 212 Emerging Trends in Managing Drinking Water – Case Studies of Coastal Villages in Karnataka Manasi S, Latha N and K V Raju
- 213 Spatio-Temporal Analysis of Forests Under Different Management Regimes Using Landsat and IRS Images Sunil Nautiyal
- 214 Traditional Knowledge System (Medicine): A Case Study of Arakalgud Taluk, Karnataka, India B K Harish, K Lenin Babu
- 215 Tribal Movement in Orissa: A Struggle Against Modernisation? Patibandla Srikant
- 216 Technological Progress, Scale Effect and Total Factor Productivity Growth in Indian Cement Industry: Panel Estimation of Stochastic Production Frontier Sabuj Kumar Mandal and S Madheswaran
- 217 Fisheries and Livelihoods in Tungabhadra Basin, India: Current Status and Future Possibilities Manasi S, Latha N and K V Raju
- 218 Economics of Shrimp Farming: A Comparative Study of Traditional Vs. Scientific Shrimp Farming in West Bengal Poulomi Bhattacharya
- 219 Output and Input Efficiency of Manufacturing Firms in India: A Case of the Indian Pharmaceutical Sector Mainak Mazumdar, Meenakshi Rajeev and Subhash C Ray

- 220 Panchayats, Hariyali Guidelines and Watershed Development: Lessons from Karnataka N Sivanna
- 221 Gender Differential in Disease Burden: It's Role to Explain Gender Differential in Mortality Biplab Dhak and Mutharayappa R
- 222 Sanitation Strategies in Karnataka: A Review
- Veerashekharappa and Shashanka Bhide 223 A Comparative Analysis of Efficiency and
- productivity of the Indian Pharmaceutical Firms: A Malmquist-Meta-Frontier Approach
  - Mainak Mazumdar and Meenakshi Rajeev
- 224 Local Governance, Patronage and Accountability in Karnataka and Kerala Anand Inbanathan
- 225 Downward Dividends of Groundwater Irrigation in Hard Rock Areas of Southern Peninsular India Anantha K H
- 226 Trends and Patterns of Private Investment in India Jagannath Mallick
- 227 Environmental Efficiency of the Indian Cement Industry: An Interstate Analysis Sabuj Kumar Mandal and S Madheswaran
- 228 Determinants of Living Arrangements of Elderly in Orissa: An Analysis Akshaya Kumar Panigrahi
- 229 Fiscal Empowerment of Panchayats in India: Real or Rhetoric? M Devendra Babu
- 230 Energy Use Efficiency in Indian Cement Industry: Application of Data Envelopment Analysis and Directional Distance Function Sabuj Kumar Mandaland S Madheswaran
- 231 Ethnicity, Caste and Community in a Disaster Prone Area of Orissa Priya Gupta
- 232 Koodankulam Anti-Nuclear Movement: A Struggle for Alternative Development? Patibandla Srikant
- 233 History Revisited: Narratives on Political and Constitutional Changes in Kashmir (1947-1990) Khalid Wasim Hassan

Price: Rs. 30.00



# INSTITUTE FOR SOCIAL AND ECONOMIC CHANGE

Dr V K R V Rao Road, Nagarabhavi P.O., Bangalore - 560 072, India Phone: 0091-80-23215468, 23215519, 23215592; Fax: 0091-80-23217008 E-mail: lekha@isec.ac.in; Web: www.isec.ac.in

ISBN 81-7791-190-2